Author's response to reviews

Title: Investigation of Swedish cases reveals an outbreak of cryptosporidiosis at a Norwegian hotel with possible links to in-house water systems

Authors:

Agnes Hajdu (agneshajdu@yahoo.com)
Line Vold (line.vold@fhi.no)
Torild A Ostmo (torild.agnalt.ostmo@mattilsynet.no)
Anna Helleve (anna.helleve@mattilsynet.no)
Sigrid R Helgebostad (sigrid.ryg.helgebostad@asker.kommune.no)
Truls Krogh (truls.krogh@fhi.no)
Lucy Robertson (lucy.robertson@veths.no)
Birgitta de Jong (birgitta.dejong@sll.se)
Karin Nygard (karin.nygard@fhi.no)

Version: 3 Date: 23 October 2008

Author's response to reviews:

Major formatting changes requested: Now each affiliation is written in full, indicating the department (when applicable), institute, city, and country.

Minor formatting changes requested: We formatted and cropped the figure according to the online guide. We changed "figure" to "figure 1" in the text.

Furthermore, we have checked and corrected the manuscript for typographical errors noticed.